%  \documentclass[10pt]{article}

\documentclass[finalversion]{FPSAC2021}
\articlenumber{50}





















%=======================================================
% usepackage
\usepackage{latexsym}
\usepackage{amssymb}
\usepackage{amsthm}
\usepackage{amscd}
\usepackage{amsmath}
\usepackage{mathrsfs}
\usepackage{graphicx}
\usepackage{hyperref}
\usepackage{shuffle}
%\usepackage{mathabx}
\usepackage{tikz-cd}
\usepackage[all]{xy}
\input xy \xyoption{frame}
\usepackage{ytableau}
\usepackage{mathtools}

\usepackage[colorinlistoftodos]{todonotes}
\usetikzlibrary{chains,scopes,decorations.markings}

%=========================================================================
% theorem declarations
\numberwithin{equation}{section}

\theoremstyle{definition}
\newtheorem* {theorem*}{Theorem}
\newtheorem* {conjecture*}{Conjecture}
\newtheorem{theorem}{Theorem}[section]
\newtheorem{thmdef}[theorem]{Theorem-Definition}
\newtheorem{propdef}[theorem]{Proposition-Definition}
\newtheorem{question}[theorem]{Question}
\theoremstyle{definition}
\newtheorem{hypothesis}[theorem]{Hypothesis}
\theoremstyle{definition}
\newtheorem{observation}[theorem]{Observation}
\newtheorem* {claim}{Claim}
\newtheorem* {example*}{Example}
\newtheorem* {fact}{Fact}
\newtheorem{lemma}[theorem]{Lemma}
\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\theoremstyle{definition}
\newtheorem* {notation}{Notation}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{algorithm}[theorem]{Algorithm}
\newtheorem* {remark*}{Remark}
\newtheorem {remark}[theorem]{Remark}
\theoremstyle{definition}
\newtheorem {example}[theorem]{Example}
\theoremstyle{definition}
\newtheorem* {examples}{Examples}
\theoremstyle{definition}
\newtheorem{exercise}[theorem]{Exercise}
\theoremstyle{definition}
\newtheorem{problem}[theorem]{Problem}
\theoremstyle{definition}
\newtheorem* {remarks}{Remarks}



\def\rW{]}
\def\lW{[}

\def\modu{\ (\mathrm{mod}\ }
\def\wh{\widehat}
\def\({\left(}
\def\){\right)}
\newcommand{\sP}{\mathscr{P}}
\newcommand{\FF}{\mathbb{F}}
\newcommand{\CC}{\mathbb{C}}
\newcommand{\QQ}{\mathbb{Q}}
\newcommand{\cP}{\mathcal{P}}
\newcommand{\cQ}{\mathcal{Q}}
\newcommand{\cK}{\mathcal{K}}
\newcommand{\cO}{\mathcal{O}}
\newcommand{\cR}{\mathcal{R}}
\newcommand{\V}{\mathcal{V}}
\newcommand{\cS}{\mathcal{S}}
\newcommand{\cI}{\mathcal{I}}
\newcommand{\cJ}{\mathcal{J}}
\newcommand{\cC}{\mathcal{C}}
\newcommand{\cD}{\mathcal{D}}
\newcommand{\cZ}{\mathcal{Z}}
\def\cS{\mathcal{S}}
\def\cX{\mathcal{X}}
\def\cW{\mathcal{W}}
\def\cY{\mathcal{Y}}
\def\NN{\mathbb{N}}
\def\tr{\mathrm{tr}}
\def\Hom{\mathrm{Hom}}
\def\End{\mathrm{End}}
\def\CC{\mathbb{C}}
\def\RR{\mathbb{R}}
\def\ZZ{\mathbb{Z}}
\def\Aut{\mathrm{Aut}}
\def\X{\mathfrak{X}}
\def\sl{\mathfrak{sl}}
\def\Ind{\mathrm{Ind}}
\def\GL{\mathrm{GL}}
\def\ind{\mathrm{ind}}
\def\Res{\mathrm{Res}}
\def\Inf{\mathrm{Inf}}
\def\ch{\mathrm{ch}}
\def\spanning{\textnormal{-span}} 
\def\Irr{\mathrm{Irr}}
\def\wt{\widetilde}

%\newcommand\wt{\operatorname{wt}}

\def\sh{\mathrm{sh}}
\def\diag{\mathrm{diag}}
\def\ts{\textstyle}
\newcommand{\fkn}{\mathfrak{n}}
\newcommand{\supp}{\mathrm{supp}}
\newcommand{\cM}{\mathcal{M}}
\newcommand{\cN}{\mathcal{N}}
\newcommand{\cL}{\mathcal{L}}
\newcommand{\g}{\mathfrak{g}}
\newcommand{\One}{{1\hspace{-.15.5cm} 1}}
\newcommand{\one}{{1\hspace{-.11cm} 1}}
\def\X{\mathcal{X}}
\newcommand{\sgn}{\mathrm{sgn}}
\newcommand{\fka}{\mathfrak{a}}
\newcommand{\fkh}{\mathfrak{h}}

\def\FC{\operatorname{FC}}

\def\fk{\mathfrak}
\def\fkm{\mathfrak{m}}

\def\barr{\begin{array}}
\def\earr{\end{array}}
\def\ba{\begin{aligned}}
\def\ea{\end{aligned}}
\def\be{\begin{equation}}
\def\ee{\end{equation}}

\def\Cyc{\mathrm{Cyc}}
\def\Fix{\mathrm{Fix}}

\def\qquand{\qquad\text{and}\qquad}
\def\quand{\quad\text{and}\quad}
\def\qquord{\qquad\text{or}\qquad}
\def\quord{\quad\text{or}\quad}

\newcommand{\gl}{\mathfrak{gl}}
\newcommand{\Sym}{\operatorname{Sym}}

\newcommand{\im}{\operatorname{im}}

\def\inv{\mathrm{Inv}}

\def\I{\mathcal{I}}
\def\M{\mathcal{M}}
\def\cH{\mathcal H}
\def\cM{\mathcal M}
\def\DesR{\mathrm{Des}_R}
\def\DesL{\mathrm{Des}_L}
\def\DesV{\mathrm{Des}_V}

\def\omdef{\overset{\mathrm{def}}}
\def\uS{\underline S}
\def\hs{\hspace{0.5mm}}

\def\uW{\underline W}
\def\uw{\underline w}

\def\RIGHT{\{\to\}}
\def\LEFT{\{\leftarrow\}}

\def\ds{\displaystyle}

\def\proj{\mathrm{proj}}

\def\id{\mathrm{id}}
\def\PP{\mathbb{P}}
\def\dash{\hs\text{---}\hs}

\def\fkS{\fk S}
\def\tphi{\tilde \phi}
\def\tpsi{\tilde \psi}


\def\ben{\begin{enumerate}}
\def\een{\end{enumerate}}

\def\cT{\mathscr{T}}
\def\sU{\mathscr{U}}
\def\cE{\mathcal E}
\def\cF{\mathcal F}
\def\cV{\mathcal V}
\def\invol{\textbf{invol}}

\newcommand{\Mon}[1]{\mathscr{F}(#1)}


\def\uL{ \underline{L}}
\def\uR{R}

\def\hs{\hspace{0.5mm}}
\def\simstar{{\hs\hs\sim_{\I_*}\hs\hs}}


\def\fpf{{\tt {FPF}}}

\def\D{\hat D}
\newcommand{\wfpf}{\Theta}
\def\DesF{\mathrm{Des}_\fpf}
\def\DesI{\mathrm{Des}_I}
\def\DesIF{\mathrm{Des}_{F}}
\def\Dfpf{\hat D_{\fpf}}
\def\cfpf{\hat c_{\fpf}}
\def\ilambda{\mu}
\def\flambda{\nu}

\def\Des{\mathrm{Des}}

\def\ellhat{\hat\ell}
\def\ellfpf{\ellhat_\fpf}
\def\Ffpf{\hat F^\fpf}
\def\Sfpf{\hat {\fk S}^\fpf}
\def\ifpf{\iota}

\def\Ifpf{\I_\fpf}

\def\x{\textbf{x}}
\def\y{\textbf{y}}
\def\z{\textbf{z}}
\def\a{\textbf{a}}
\def\b{\textbf{b}}
\def\c{\textbf{c}}
\def\d{\textbf{d}}
\def\e{\textbf{e}}


%\newcommand{\xRightarrow}[2][]{\ext@arrow 0359\Rightarrowfill@{#1}{#2}}

\newcommand{\Fl}{\operatorname{Fl}}
\newcommand{\std}{\operatorname{std}}
\newcommand{\Proj}{\operatorname{Proj}}
\newcommand{\dist}{\operatorname{dist}}

\renewcommand{\O}{\operatorname{O}}
\newcommand{\Sp}{\operatorname{Sp}}
\newcommand{\stb}{\operatorname{stb}}
\newcommand{\Ess}{\operatorname{Ess}}
\newcommand{\rank}{\operatorname{rank}}
\renewcommand{\dim}{\operatorname{dim}}

\renewcommand{\r}[1]{\textcolor{red}{#1}}
\renewcommand{\skew}{\operatorname{skew}}
\newcommand{\rk}{\operatorname{rk}}

\def\dbldots{: \hspace{.55mm} : \hspace{.55mm} :}

\def\L{\underline L}
\def\R{R}

\newcommand{\cA}{\mathcal{A}}
\newcommand{\cB}{\mathcal{B}}
\newcommand{\iB}{\hat\cB}
\newcommand{\fB}{\hat\cB^\fpf}

\def\HAfpf{\HA^\fpf}
\def\cAfpf{\cA^\fpf}
\def\cRfpf{\hat\cR_\fpf}
\def\iR{\hat\cR}
\def\iHR{\cH\hat\cR}

\def\act{\rtimes}
\newcommand\dact{\hat\circ}
\def\dem{\circ}

\def\invdemspace{\hspace{.85mm}}
\newcommand{\geninvdem}[3]{#2 \invdemspace \dact#1\invdemspace #3}
\newcommand{\idem}[2]{\geninvdem{_*}{#1}{#2}}
\newcommand{\sidem}[2]{\geninvdem{}{#1}{#2}}

\def\ellhat{\hat\ell}
\def\Ffpf{\hat F^\fpf}
\def\Sfpf{\hat {\fk S}^\fpf}
\def\iS{\hat \fkS}

\def\Luc{\mathrm{Luc}}

\newcommand{\ct}{\operatorname{ct}}

\newcommand{\del}{\operatorname{del}}

\newcommand{\push}{\mathord{\downarrow}}
\newcommand{\pull}{\mathord{\uparrow}}

\newcommand{\ipush}{\mathord{\Downarrow}}
\newcommand{\ipull}{\mathord{\Uparrow}}

\def\sM{\mathscr{M}}
\def\ttS{{\tt S}}

\def\s{\mathbf{s}}
\def\t{\mathbf{t}}
\def\r{\mathbf{r}}
\def\customstar{\hs\hat \bullet_{\mathrm{des}}\hs}

\def\red{\emph{reduce}}
\def\sA{\mathscr{A}}
\def\sB{\mathscr{B}}
\def\sR{\mathscr{R}}
\def\sC{\mathscr{C}}
\def\sE{\mathscr{E}}
\def\ihB{\underline{\isB}}

\def\sI{\mathscr{I}}
\def\tS{\tilde S}
\def\tX{G}


\def\tI{\tilde I}
\def\cG{\mathcal{G}}


\newcommand{\sAq}[1]{\sA_{\bullet,\geq #1}}
\newcommand{\sAp}[1]{\sA_{\leq #1,\bullet}}

\def\sX{\mathscr{X}}
\def\sY{\mathscr{Y}}

\def\u{\mathbf{u}}
\def\v{\mathbf{v}}
\def\isB{\hat{\mathscr{B}}}

\def\cWmin{\cM}

\def\UT{\operatorname{UT}}
\def\traj{\operatorname{traj}}


\def\Inv{\operatorname{Inv}}
\def\HA{\cA_{\textsf{Hecke}}}
\def\iF{\hat{F}}

\def\F{\tilde F}
\def\tF{F}
\def\QSym{\textsf{QSym}}
\def\Sym{\textsf{Sym}}
\def\zetaq{\zeta_{\QSym}}
\def\Par{\textsf{Par}}
\def\tI{\tilde I}
\def\iPhi{\Upsilon}
\def\fPhi{\Psi}

\def\Refl{\textsf{Ref}}
\def\Cov{\textsf{Cov}}
\def\dotequals{\mathbin{\overset{_\bullet} =}}
\def\V{\textsf{Vir}}

\newcommand{\arc}[2]{ \ar @/^#1pc/ @{-} [#2] }
\def\arcstop{\endxy\ }
\def\arcstart{\ \xy<0cm,-.06cm>\xymatrix@R=.1cm@C=.2cm }
\newcommand{\arcstartc}[1]{\ \xy<0cm,-.15cm>\xymatrix@R=.1cm@C=#1cm}

\def\diagramAB{
\arcstart
{
*{\circ}     & *{\circ}
}
\arcstop}

\def\diagramBA{
\arcstart
{
*{\circ}   \arc{.5}{r}  & *{\circ}
}

\arcstop
}
\def\diagrambAC{
\arcstart
{
*{\bullet}    \arc{.5}{r}  & *{\circ} & *{\circ}
}
\arcstop
}
\def\diagramBaC{
\arcstart
{
*{\circ}    \arc{.5}{r}  & *{\bullet} & *{\circ}
}
\arcstop
}
\def\diagramcBA{
\arcstart
{
*{\bullet}    \arc{.5}{rr}  & *{\circ} & *{\circ}
}
\arcstop
}




\def\diagramACb{
\arcstart
{
*{\circ}     & *{\circ}  \arc{.5}{r} & *{\bullet}
}
\arcstop
}
\def\diagramAcB{
\arcstart
{
*{\circ}     & *{\bullet}  \arc{.5}{r} & *{\circ}
}
\arcstop
}
\def\diagramCbA{
\arcstart
{
*{\circ}     \arc{.5}{rr}   & *{\bullet} & *{\circ}
}
\arcstop
}
\def\diagramCBa{
\arcstart
{
*{\circ}     \arc{.5}{rr}   & *{\circ} & *{\bullet}
}
\arcstop
}


\def\diagramBAdc{
\arcstart
{
*{\circ}  \arc{.5}{r}   & *{\circ}    & *{\bullet} \arc{.5}{r}& *{\bullet}
}
\arcstop
}
\def\diagramBaDc{
\arcstart
{
*{\circ}  \arc{.5}{r}   & *{\bullet}    & *{\circ} \arc{.5}{r}& *{\bullet}
}
\arcstop
}
\def\diagramBadC{
\arcstart
{
*{\circ}  \arc{.5}{r}   & *{\bullet}    & *{\bullet} \arc{.5}{r}& *{\circ}
}
\arcstop
}
\def\diagrambAdC{
\arcstart
{
*{\bullet}  \arc{.5}{r}   & *{\circ}    & *{\bullet} \arc{.5}{r}& *{\circ}
}
\arcstop
}
\def\diagrambADc{
\arcstart
{
*{\bullet}  \arc{.5}{r}   & *{\circ}    & *{\circ} \arc{.5}{r}& *{\bullet}
}
\arcstop
}
\def\diagrambaDC{
\arcstart
{
*{\bullet}  \arc{.5}{r}   & *{\bullet}    & *{\circ} \arc{.5}{r}& *{\circ}
}
\arcstop
}


\def\diagramCDab{
\arcstart
{
*{\circ}  \arc{.5}{rr}   & *{\circ} \arc{.5}{rr}   & *{\bullet} & *{\bullet}
}
\arcstop
}
\def\diagramCdAb{
\arcstart
{
*{\circ}  \arc{.5}{rr}   & *{\bullet} \arc{.5}{rr}   & *{\circ} & *{\bullet}
}
\arcstop
}
\def\diagramCdaB{
\arcstart
{
*{\circ}  \arc{.5}{rr}   & *{\bullet} \arc{.5}{rr}   & *{\bullet} & *{\circ}
}
\arcstop
}
\def\diagramcDAb{
\arcstart
{
*{\bullet}  \arc{.5}{rr}   & *{\circ} \arc{.5}{rr}   & *{\circ} & *{\bullet}
}
\arcstop
}
\def\diagramcDaB{
\arcstart
{
*{\bullet}  \arc{.5}{rr}   & *{\circ} \arc{.5}{rr}   & *{\bullet} & *{\circ}
}
\arcstop
}
\def\diagramcdAB{
\arcstart
{
*{\bullet}  \arc{.5}{rr}   & *{\bullet} \arc{.5}{rr}   & *{\circ} & *{\circ}
}
\arcstop
}

\def\diagramDbcA{
\arcstart
{
*{\circ}  \arc{.5}{rrr}   & *{\bullet}   & *{\bullet} & *{\circ}
}
\arcstop
}
\def\diagramDbCa{
\arcstart
{
*{\circ}  \arc{.5}{rrr}   & *{\bullet}   & *{\circ} & *{\bullet}
}
\arcstop
}
\def\diagramdBcA{
\arcstart
{
*{\bullet}  \arc{.5}{rrr}   & *{\circ}   & *{\bullet} & *{\circ}
}
\arcstop
}


\def\diagramDcBa{
\arcstart
{
*{\circ}  \arc{.5}{rrr}   & *{\bullet} \arc{.25}{r}   & *{\circ} & *{\bullet}
}
\arcstop
}

\def\diagramdCbA{
\arcstart
{
*{\bullet}  \arc{.5}{rrr}   & *{\circ} \arc{.25}{r}   & *{\bullet} & *{\circ}
}
\arcstop
}

\def\diagramDCba{
\arcstart
{
*{\circ}  \arc{.5}{rrr}   & *{\circ} \arc{.25}{r}   & *{\bullet} & *{\bullet}
}
\arcstop
}
\def\diagramDcbA{
\arcstart
{
*{\circ}  \arc{.5}{rrr}   & *{\bullet} \arc{.25}{r}   & *{\bullet} & *{\circ}
}
\arcstop
}
\def\diagramdcBA{
\arcstart
{
*{\bullet}  \arc{.5}{rrr}   & *{\bullet} \arc{.25}{r}   & *{\circ} & *{\circ}
}
\arcstop
}





\def\tp{\Theta^+}
\def\tm{\Theta^-}
\def\tpm{\{\Theta^\pm\}}
\def\cF{\mathcal F}
\def\cFp{\cF_n^+}
\def\cFm{\cF_n^-}
\def\h {\mathrm{ht}}
\def\Cyc{\mathrm{Cyc}}
\def\wCyc{\widetilde\Cyc}
\def\sfpf{\sgn_\fpf}
\def\m{\mathfrak{m}}
\def\n{\mathfrak{n}}
\def\Em{E_\m}
\def\En{E_\n}
\def\E{E}
\def\invsim_i{\overset{\mathrm{i}}{\underset{\mathrm{inv}}{\sim}}}



\def\LL{\QQ[x,x^{-1}]}
\def\H{\mathcal{H}}
\def\A{\mathsf{A}}
\def\BC{\mathsf{BC}}
\def\D{\mathsf{D}}
%\def\one{M}
\def\Ifpf{I^{\mathsf{FPF}}}
\def\ellfpf{\ell^{\mathsf{FPF}}}
\def\AbsFix{\mathsf{Fix}^\pm}
\def\cG{\mathcal{G}}
\def\G{\mathbf{G}}
\def\Asc{\mathrm{Asc}}
\def\cP{\mathscr{P}}

\def\WB{W^{\mathsf{BC}}}
\def\WD{W^{\mathsf{D}}}

\def\cFB{\cF^{\mathsf{BC}}}
\def\cFD{\cF^{\mathsf{D}}}

\def\cKB{\cK^{\mathsf{BC}}}
\def\cKD{\cK^{\mathsf{D}}}

\def\SB{S^{\mathsf{BC}}}
\def\SD{S^{\mathsf{D}}}

\def\m{\mathbf{m}}
\def\n{\mathbf{n}}

\def\fromRSK{\xleftarrow{\hs\mathsf{RSK}\hs}}
\def\row{{\mathfrak{row}}}
\def\RSK{\xrightarrow{\mathsf{RSK}}}
\def\PRSK{P_{\mathsf{RSK}}}
\def\QRSK{Q_{\mathsf{RSK}}}

\def\PMRSK{P_{\fk m}}
\def\PNRSK{P_{\fk n}}

\def\fromMRSK{\xleftarrow{\hs\fkm\hs}}
\def\fromNRSK{\xleftarrow{\hs\fkn\hs}}
\def\toNRSK{\xrightarrow{\hs\fkn\hs}}
\def\toNRSK{\xrightarrow{\hs\fkn\hs}}

\newcommand{\ytab}[1]{
\ytableausetup{boxsize = .4cm,aligntableaux=center}
 \begin{ytableau} #1 \end{ytableau}}


\def\Ad{\mathrm{Ad}}
\def\W{W^{\BC}}
\def\WD{W^{\D}}

\def\TT{\mathbb{T}}


\usepackage[colorinlistoftodos]{todonotes}
\newcommand{\eric}[1]{\todo[size=\tiny,inline,color=green!30]{#1 \\ \hfill --- E.}}
\newcommand{\yifeng}[1]{\todo[size=\tiny,inline,color=blue!30]{#1 \\ \hfill --- Y.}}









\title{Perfect models and Gelfand $W$-graphs}

\author{
Eric Marberg\thanks{\href{mailto:emarberg@ust.hk}{emarberg@ust.hk}. The first author was partially supported by RGC Grant ECS 26305218.
}\addressmark{1}
\and Yifeng Zhang\thanks{\href{mailto:yzhangci@connect.ust.hk}{yzhangci@connect.ust.hk}.
The second author was partially supported by RGC Grant ECS 26305218.}\addressmark{1}
}

%% then use \addressmark to match authors to institutions here
\address{\addressmark{1}Department of Mathematics, Hong Kong University of Science and Technology}


%\author{Eric Marberg \and Yifeng Zhang}


%% put the date of submission here
\received{December 1, 2020}

%% leave this blank until submitting a revised version
\revised{April 25, 2021}

%% put your English abstract here, or comment this out if you don't have one yet
%% please don't use custom commands in your abstract / resume, as these will be displayed online
%% likewise for citations -- please don't use \cite, and instead write out your citation as something like (author year)
\abstract{
A Gelfand model for an algebra is a module isomorphic to a direct sum of irreducible modules, 
with every isomorphism class  of irreducible modules represented exactly once.
We introduce and study the notion of a perfect model for a finite Coxeter group;
such a model is a certain set of discrete data parametrizing a Gelfand model for the associated Iwahori--Hecke algebra. We classify which Coxeter groups have perfect models,
and then describe explicit Gelfand models for the classical finite Coxeter groups. This generalizes separate constructions of Adin, Postnikov, and Roichman
and of Araujo and Bratten.
Our Gelfand models have interesting ``canonical bases'' that give rise to associated $W$-graphs. We classify the molecules in these $W$-graphs when $W$ is a symmetric group,
and conjecture that in type A every molecule is a cell.
}

%% put your French abstract here, or comment this out if you don't have one
%\resume{}

%% put your keywords here, or comment this out if you don't have them yet
\keywords{Coxeter systems, Iwahori--Hecke algebras, $W$-graphs, Gelfand models, perfect involutions, quasiparabolic sets}

%% you can include your bibliography however you want, but using an external .bib file is STRONGLY RECOMMENDED and will make the editor's life much easier
%% regardless of how you do it, please use numerical citations, ie. [xx, yy] in the text

%% this sample uses biblatex, which (among other things) takes care of URLs in a more flexible way than bibtex
%% but you can use bibtex if you want
\usepackage[backend=bibtex]{biblatex}
\addbibresource{50Marberg.bib}
%% note the \printbibliography command at the end of the file which goes with these biblatex commands


\begin{document}
\begin{thebibliography}{99}
% 
% \bibitem{APR2007} R. Adin, A. Postnikov, and Y. Roichman, Combinatorial Gelfand models, \emph{J. Algebra} \textbf{320} (2008), 1311--1325. 
% 
%  \bibitem{APR2010} R. Adin, A. Postnikov, and Y. Roichman, A Gelfand model for wreath products, 
%  \emph{Israel J. Math.} \textbf{179} (2010), 381--402.
%  
%    \bibitem{AraujoBratten} J. O. Araujo and T. Bratten, Gelfand models for classical Weyl groups,
%  \emph{J. Algebra} \textbf{403} (2014), 154--178.
%
%\bibitem{Baddeley} R. W. Baddeley, Models and involution models for wreath products and certain Weyl groups, \emph{J. London Math. Soc.} (2) \textbf{44} (1991), 55--74.
%
%\bibitem{BaddeleyThesis} R. W. Baddeley, \emph{Some Multiplicity-Free Characters of Finite Groups}, Ph.D. Thesis, Cambridge 1991. 
%
% \bibitem{Beissinger} J. S. Beissinger, Similar construtions for Young tableaux and involutions, and their application to shiftable tableaux,
%\emph{Discrete Math.} \textbf{67} (1987), 149--163.
%
%
%\bibitem{BumpGinzburg} D. Bump, D. Ginzburg, Generalized Frobenius-Schur numbers, \emph{J. Algebra} \textbf{278} (2004), 
%294--313. 
%
%
%\bibitem{Caselli2009} F. Caselli, Involutory reflection groups and their models, \emph{J. Algebra} \textbf{324} (2010), 370--393. 
%
%\bibitem{CF} F. Caselli and R. Fulci, Refined
%Gelfand models for wreath products, \emph{Europ. J. Comb.} \textbf{32} (2011), 198--216.
%
%
%\bibitem{CM} F. Caselli and E. Marberg, Isomorphisms, automorphisms, and generalized involution models of projective reflection groups,  \emph{Israel J. Math.} \textbf{199} (2014), 433--483.
%
%\bibitem{HowlettYin} R. B. Howlett and Y. Yin, Inducing $W$-Graphs,
%\emph{Math. Z.} \textbf{244} (2003) 415--431.
%
%\bibitem{HowlettYin2} R. B. Howlett and Y. Yin, Inducing $W$-graphs II, \emph{Manuscripta Mathematica} \textbf{115} (2004), no. 4, 495--511.
%
%
%\bibitem{IRS}{N. F. J. Inglis, R. W. Richardson, and J. Saxl}, An explicit model for the complex representations of $S_n$, \emph{Arch. Math. (Basel)} \textbf{54} (3) (1990) 258--259.
%
%\bibitem{KL} D. Kazhdan and G. Lusztig, Representations of Coxeter groups and Hecke algebras, \emph{Invent. Math.} \textbf{53} (1979), 165--184.
%
%\bibitem{KV2004} V. Kodiyalam and D. N. Verma, A natural representation model for symmetric groups, preprint (2004), {\tt arXiv:math/0402216v1}.
%
%\bibitem{LV} G. Lusztig and D. A. Vogan, Hecke algebras and involutions in Weyl groups, 
%\emph{Bull. Inst. Math. Acad. Sinica (N.S.)} \textbf{7} (2012), 323-354.
%
%\bibitem{M2011} E. Marberg, Automorphisms and generalized involution models of finite complex reflection groups, 
%\emph{J.  Algebra} \textbf{334} (2011), 295--320.
%
%\bibitem{M2012} E. Marberg, Generalized involution models for wreath products, \emph{Israel J. Math.} \textbf{192} (2012), 157--195.
%
%\bibitem{Marberg} E. Marberg, Bar operators for quasiparabolic conjugacy classes in a Coxeter group,
%\emph{J. Algebra} \textbf{453} (2016), 325--363.
%
%
%\bibitem{MZ} E. Marberg and Y. Zhang, Perfect models and Gelfand $W$-graphs, 
%preprint (2020), {\tt arXiv:2012.13868}.
%
%\bibitem{RV} E. M. Rains and M. J. Vazirani, Deformations of permutation representations of Coxeter groups, \emph{J. Algebr. Comb.} \textbf{37} (2013), 455--502. 
%
%\bibitem{Stembridge} J. R. Stembridge, Admissible $W$-Graphs, \emph{Representation Theory} \textbf{12} (2008), 346--368.
%
%
%\bibitem{Vinroot} C. R. Vinroot, Involution models of finite Coxeter groups, \emph{J. Group Theory} \textbf{11} (2008), 333--340.
%
%
%\end{thebibliography}

\maketitle


\setcounter{tocdepth}{2}
%\tableofcontents


 \section{Introduction}


In this article we study certain uniform ways 
of constructing multiplicity-free representations of Iwahori--Hecke algebras
that are also instances of \emph{$W$-graphs}.
We begin by explaining the definition of the models that are our primary subject.

Let $(W,S)$ be a Coxeter system with length function $\ell : W \to \NN \coloneqq \{0,1,2,\dots\}$.
Define $\Aut(W,S)$ to be the group of \emph{Coxeter automorphisms} of $W$, that is, group automorphisms $\varphi \in \Aut(W)$ with $\varphi(S) = S$.
Let $W^+ = W \rtimes \Aut(W,S)$ be the semidirect product whose elements are the pairs $(w, \varphi)$ with $w \in W$ and $\varphi \in \Aut(W,S)$, with multiplication 
\[ (v,\alpha)(w,\beta) \coloneqq (v \cdot \alpha(w), \alpha\beta).\]
Extend the length function of $W$ to $W^+$ by setting $\ell((w,\varphi)) = \ell(w)$.
We view $W$ as a subgroup of $W^+$ by identifying $w \in W$ with $(w,\id_W) \in W^+$.

Let $z = (w,\varphi) \in W^+$.
Following \cite{RV}, we define $z$ to be a \emph{perfect involution} if 
$z^2= (z t )^4 = 1 $ for all reflections
$t \in  \{ vsv^{-1} : v \in W, s \in S\}.$
Let $\sI = \sI(W,S)$ be the set of perfect involutions in $W^+$. 
The group $W$ acts on $\sI$ by conjugation
\[v : (w,\varphi) \mapsto v (w,\varphi)v^{-1} =  (v\cdot w \cdot\varphi(v)^{-1},\varphi) .\]
%
As in \cite{RV}, we say an element $z \in \sI$ is \emph{$W$-minimal}
if $\ell(sz s) \geq \ell(z)$ for all $s \in S$, and \emph{$W$-maximal}
if $\ell(sz s) \leq \ell(z)$ for all $s \in S$.
Each $W$-conjugacy class in $\sI$ contains a unique $W$-minimal element
(and a unique $W$-maximal element when $|W|<\infty$) \cite[Cor. 2.10]{RV}.


If $C_W(\theta) \coloneqq \{ w \in W: wz = z w\}$ where
$z\in \sI$ is $W$-minimal,
 then $C_W(z)$ is a \emph{quasiparabolic subgroup} of $W$ in the sense of Rains and Vazirani \cite[Thm. 4.6]{RV}. Among other consequences, this implies that the 
 permutation representation of $W$ acting on  $W/C_W(z)$ naturally deforms to a representation of the Iwahori--Hecke algebra of $(W,S)$.


\begin{example}\label{ex-1}
%Rains and Vazirani classify the perfect involutions in the three families of finite classical Coxeter groups in 
The sets $\sI(W,S)$ for classical Coxeter systems are described in
\cite[\S9]{RV}:
\ben
\item[($\A$)] The symmetric group $S_n$ of permutations of $[n] \coloneqq \{1,2,\dots,n\}$
is a Coxeter group relative to the generating set $\{s_1,s_2,\dots,s_{n-1}\}$ where $s_i\coloneqq(i,i+1)$.
The only non-identity element of $\Aut(S_n,\{s_1,s_2,\dots,s_{n-1}\})$ is the inner automorphism $\Ad(w_0)$
induced by  $w_0 = n\cdots 321$. 
When $n$ is odd every perfect involution in $S_n^+$ is central.
When $n$ is even there are three $S_n$-minimal elements in $S_n^+$: the identity element $1 \in S_n $, the fixed-point-free involution $s_1s_3s_5\cdots s_{n-1} \in S_n$, and $(1, \Ad(w_0)) \in S_n^+$.

\item[($\BC$)] The group $\W_n$ of permutations of $\{-n,\dots,-1,0,1,\dots,n\}$ that commute with the negation map
is a Coxeter group relative to the generators $\{s_0,s_1,\dots,s_{n-1}\}$ 
with $s_0 \coloneqq (-1,1)$ and $s_i \coloneqq (-i-1,-i)(i,i+1)$ for $i>0$. 
The one-line representation of $w \in \W_n$ is the word $w(1)w(2)\cdots w(n)$ where we write $\bar i$ in place of $-i$.
When $n>2$ there are only trivial Coxeter automorphisms 
and the perfect involutions in $\W_n$ are the elements $w=w^{-1}$
such that $|w| : i \mapsto |w(i)|$ is the identity map or a fixed-point-free permutation of $[n]$.
The $\W_n$-minimal perfect involutions consist of the $n+1$ elements
$\theta_i \coloneqq \bar 1 \bar 2 \cdots \bar i (i+1)\cdots n$  plus $s_1s_3s_5\cdots s_{n-1}$ if $n$ is even.

\item[($\D$)] The subgroup $\WD_n$ of $w \in \W_n$ such that $|\{ i \in [n] : w(i) < 0\}|$ is even 
is a Coxeter group relative to the generators $\{s_{-1},s_1,s_2,\dots,s_{n-1}\}$ where $s_{-1} \coloneqq (-2,1)(-1,2)$
and $s_i $ for $i\geq 0$ is as in ($\BC$).
For $n>4$ there is just one nontrivial Coxeter automorphism  $w\mapsto w^\ast \coloneqq s_0ws_0$.
The perfect involutions in $(\WD_n)^+$ are the pairs $(w, \varphi)$ with $\varphi \in \{1,\ast\}$ and $w^{-1} = \varphi(w)$ such that $|w| \in S_n$ is $1$ or fixed-point-free.
The $\WD_n$-minimal perfect involutions are 
$\theta_i$ for $i$ even, $\tilde \theta_i \coloneqq (1 \bar2 \bar3 \cdots \bar i (i+1)\cdots n, \ast)$ for $i$ odd,   plus both $s_1s_3s_5\cdots s_{n-1}$ and $s_{-1}s_3s_5\cdots s_{n-1}$ if $n$ is even.
\een
\end{example}







Given  $J\subset S$, let $W_J \coloneqq \langle s \in J\rangle$. % be the corresponding parabolic subgroup. 
Then $(W_J, J)$ is a Coxeter system
whose length function is $\ell|_{W_J}$.
Write $\sI_J \coloneqq \sI(W_J,J)$ for the set of perfect involutions in $(W_J)^+$.

\begin{definition}
A \emph{model triple} $(J, z_{\min}, \sigma)$ for a finite Coxeter group $W$ consists of
a set $J\subset S$, a $W_J$-minimal element $z_{\min} \in  \sI_J$,
and a linear character $\sigma : W_J \to \{\pm 1\}$.
A set $\sP$ of model triples for $W$ is a \emph{perfect model}   if
$ \sum_{(J,z_{\min},\sigma) \in \sP} \Ind_{C_{W_J}(z_{\min})}^W \Res^{W_J}_{C_{W_J}(z_{\min})}(\sigma) = \sum_{\chi \in \Irr(W)} \chi$,
where $\Irr(W)$ is the set of complex irreducible characters of $W$
and $\Ind_H^G$ and $\Res_H^G$ denote the usual operations of induction and restriction
for groups $H\subset G$.
\end{definition}

Our first objective is to classify which finite Coxeter groups have perfect models. We will then investigate how to use such models to construct some interesting representations 
of the Iwahori--Hecke algebras of finite classical Coxter systems.
Since space is limited, we have omitted all proofs in this extended abstract; for 
these details, see \cite{MZ}.

\section{Classification of perfect models}

A Coxeter system $(W,S)$ is \emph{irreducible} if the only disjoint decomposition $S=S'\sqcup S''$ 
in which every $s' \in S'$ commutes with every $s'' \in S''$ has either $S'=\varnothing\neq S''$ or $S' \neq \varnothing=S''$.
An \emph{irreducible factor} of $(W,S)$ is a parabolic subsystem $(W_J,J)$ that is irreducible.

\begin{theorem}\label{main-thm1}
A finite Coxeter group $W$ 
has a perfect model if and only if each of its irreducible factors has a perfect model.
The irreducible Coxeter systems
with perfect models are those of type $\A_{n-1}$, $\BC_n$,
$\D_{2n+1}$, $\mathsf{H}_3$, and $\mathsf{I}_2(n)$ for all $n\geq 2$.
The irreducible Coxeter systems without perfect models are
those of type $\D_{2n}$, $\E_6, \E_7, \E_8$, $\mathsf{F}_4$, and $\mathsf{H}_4$.
\end{theorem}

Perfect models for the finite classical Coxeter groups are given below.
The model for type $\A_{n-1}$ is well-known; see \cite{APR2007,IRS,KV2004}.
What we describe for type $\BC_n$ (respectively, $\D_n$)
is a coarser version of the models in
\cite{APR2010,Baddeley,M2012}
(respectively, \cite{Caselli2009,CF, CM,M2011}).


\begin{definition}\label{main-thm1-def}
 Let $W \in \{ S_n, \W_n, \WD_n\}$ be a classical Weyl group.
 Write $\one=\one_W : w \mapsto 1$ and $\sgn=\sgn_W : w \mapsto (-1)^{\ell(w)}$ for the trivial and sign 
 representations.
%of type $\A$, $\BC$, or $\D$.
\ben

\item[($\A$)] Assume $W=S_n$ with $n\geq 1$.  For each integer $0\leq k\leq \lfloor\frac{n}{2}\rfloor$, form a triple $(J,z_{\min},\sigma)$ 
by taking $J = \{s_1,s_2,\dots,s_{n-1}\} \setminus\{s_{2k}\}$
so that  $ \langle J\rangle  = S_{2k} \times S_{n-2k} $,
and then setting
\[z_{\min} = s_1s_3s_5\cdots s_{2k-1}
\quand \sigma = \one_{S_{2k}} \times \sgn_{S_{n-2k}}.\]



\item[($\BC$)] Assume $W=\W_n$ with $n\geq 2$.  For each $0\leq k\leq \lfloor\frac{n}{2}\rfloor$, form a triple
  $(J,z_{\min},\sigma)$ by taking
$J = \{s_0,s_1,s_2,\dots,s_{n-1}\} \setminus\{s_{2k}\}$
 so that  $ \langle J\rangle  = \W_{2k} \times S_{n-2k}  $,
 and then setting
\[z_{\min} = s_1s_3s_5\cdots s_{2k-1}
\quand \sigma = \one_{\W_{2k}} \times \sgn_{S_{n-2k}}.\]

\item[($\D$)] Assume $W=\WD_n$ with $n\geq 3$. For each $0<k\leq \lfloor\frac{n}{2}\rfloor$, form a triple
  $(J,z_{\min},\sigma)$ by taking
$J = \{s_{-1},s_1,s_2,\dots,s_{n-1}\} \setminus\{s_{2k}\}$
 so that $ \langle J\rangle  = \WD_{2k} \times S_{n-2k}  $,
 and then setting
\[z_{\min} = s_1s_3s_5\cdots s_{2k-1}
\quand\sigma = \one_{\WD_{2k}} \times \sgn_{S_{n-2k}}.\]
Also include one additional triple $(J,z_{\min},\sigma)=(\{s_1,s_2,\dots,s_{n-1}\}, 1, \sgn_{S_n})$.

\een
Let $\sP=\sP(W)$ be the set of
 $1+\lfloor \frac{n}{2}\rfloor$ triples $(J,z_{\min},\sigma)$ listed in each case.
 \end{definition}
 


\begin{theorem}
\label{main-thm1b}
Assume $W$ is one of the classical Weyl groups $S_{n-1}$, $\W_n$, or $\WD_{2n+1}$ for an integer $n\geq 2$. 
Then the set  $\sP(W)$ given in Definition~\ref{main-thm1-def} is a perfect model.
\end{theorem}


Let $G$ be a finite group. Choose representatives for the distinct conjugacy classes  
of involutions $g=g^{-1} \in G$. For each representative involution $g$, form the centralizer $H = C_G(g) = \{x \in G : xg = gx\}$
and choose a linear character $\lambda : H \to \CC$.
The resulting set of pairs $(H,\lambda)$ is 
an \emph{involution model} for $G$ if $\sum_{(H,\lambda)} \Ind_H^G(\lambda) = \sum_{\chi \in \Irr(G)} \chi$.  
The existence of an involution model implies
that $ |\{ g \in G : g=g^{-1}\}| = \sum_{\chi \in \Irr(G)} \deg \chi$, which holds if and only if all representations of $G$ are realizable over $\RR$ (see \cite[\S2]{BumpGinzburg}).

The involution models for finite Coxeter groups
are classified in \cite{BaddeleyThesis,Vinroot}.
Surprisingly, although 
we do not know of any general procedure for converting
involution models to perfect models or vice versa,
this classification is identical to the one in
Theorem~\ref{main-thm1}:

\begin{corollary}
A finite Coxeter group has a perfect model if and only if it has an involution model.
\end{corollary}




\section{Gelfand models for Iwahori--Hecke algebras}

Let $x$ be an indeterminate.
The \emph{(single-parameter) Iwahori--Hecke algebra} of a Coxeter system $(W,S)$ is
the free $\QQ[x,x^{-1}]$-module $\H = \H(W)$ with basis $\{H_w : w \in W\}$, equipped with the unique algebra
structure in which $H_wH_s = H_{sw}$ if $\ell(sw) > \ell(w)$
and $H_wH_s = H_{sw} +(x-x^{-1})H_w$ if $\ell(sw) < \ell(w)$ for all $s \in S$ and $w \in W$.
%\[ H_s H_w  =\begin{cases} H_{sw} &\text{if }\ell(sw) > \ell(w) \\
%H_{sw} + (x-x^{-1}) H_w &\text{if }\ell(sw) <\ell(w)
%\end{cases}
%\quad\text{for all $s \in S$ and $w \in W$.} \]

A \emph{Gelfand model} for $\H$ is an $\H$-module
 isomorphic to the direct sum of all irreducible $\H$-modules.
Any perfect model for $W$ gives rise to  
 a Gelfand model for $\H$.
Our results in this section are formal consequences of this fact and
Theorem~\ref{main-thm1b}.

 

For integers $i > 0$, define permutations 
$
s_i \coloneqq (-i-1,-i)(i,i+1),
$
$
s_0 \coloneqq (-1,1),
$
and
$ s_{-i} = (-i-1,i)(-i,i+1).
$
We realize the Coxeter groups of type $\A_{n-1}$, $\BC_n$, and $\D_n$
as 
$
S_n \coloneqq \langle s_1,s_2,\dots, s_{n-1}\rangle
\subset \WD_n \coloneqq \langle s_{-1},s_1,s_2,\dots,s_{n-1}\rangle
\subset \W_n \coloneqq \langle s_0,s_1,s_2,\dots,s_{n-1}\rangle
.$
Assume below that $W$  is one of these  groups
with $S$ the set of simple generators just listed.
Then the length function of $W$ is determined by the following properties:
\begin{itemize} 
\item If $w\in W$  and $i >0$ 
then $\ell(ws_i) < \ell(w) $ if and only if $w(i) > w(i+1)$.

\item If $w \in W=\W_n$ then $\ell(ws_0) <\ell(w) $ if and only if $w(1) < 0$.

\item If $w \in W= \WD_n$ then $\ell(ws_{-1}) < \ell(w) $ if and only if 
$-w(2) > w(1)$.
% we have either
%$w(1) <0>w(2)$ or $ 0<w(2)<-w(1) $ or $ 0 < w(1) < -w(2).$
\end{itemize}
%
Let $\cF_{2n} \coloneqq \left\{ z \in \W_{2n} : z=z^{-1}\text{ and }|z(i)| \neq i \text{ for all }i \in [2n]\right\}\subset \WD_{2n}.$
We say that an integer $i>0$ is a \emph{visible descent} of $z \in \cF_{2n}$
if $z(i+1) < \min\{i,z(i)\}$ or $ z(i) < -i.$ Next:
\begin{itemize}
\item Define $\cK_{n}^\BC$ to be the set of $z \in \cF_{2n}$ with no visible descents greater than $n$.
\item Define $\cK^\A_{n-1} \coloneqq \cK_n^\BC \cap S_{2n}$.
\item Define $\cK^\D_n$ to  be the subset of 
 $z \in \cK_n^\BC$ for which $|\{ i\in [n] : z(i) < -i\}|$
is even.
\end{itemize}
%
Define $\cK = \cK(W)$ to be $\cK^\A_{n-1}$, $\cK_n^\BC$, or $\cK_n^\D$
according to whether $W$ is $S_n$, $\W_n$, or $\WD_n$.
When evaluating $\ell(z)$ for $z \in \cK$, we consider
$
\cK^\A_{n-1} \subset S_{2n},
$
$
\cK^\BC_{n} \subset \W_{2n} ,
$
and
$
\cK^\D_{n} \subset \WD_{2n}.
$
\begin{remark}\label{notation-rmk}
One can characterize the elements of $\cK$ 
more explicitly. 
Namely,  $z \in \cF_{2n}$ belongs to $ \cK_n^\BC$
if and only if 
there is an integer $0 \leq i \leq n$ with 
$ i \equiv n \modu 2)$ such that
\[
-n \leq z(n+1) < z(n+2) < \dots < z(n+i) \leq n
\quad\text{and}\quad
z(n+i+2j) =n+i+2j-1\] 
for 
%
each 
 integer $j>0$ with $n+i+2j \leq 2n$. 
This suggests the following notation.
Suppose $w =w^{-1} \in W^\BC_n$. Let 
$a_1 >a_2  > \dots >a_p$
(respectively, $ b_1<b_2<\dots<b_q$)
be the numbers $a \in [n]$ with $w(a) = -a$ (respectively, $b \in [n]$ with $w(b) = b$).
Define $\underline w \in \cK^\BC_n$ to be the unique element mapping 
$a_i \mapsto -n -i$ and $b_i \mapsto n + p + i$ and $c \mapsto w(c)$ for $c \in [n] \setminus \{ a_1,a_2,\dots,a_p, b_1,b_2,\dots,b_q\}$.
For example, we would have
\[
\underline{2134} = 21563487 \in \cK^\A_3
\quand 
\underline{\bar 3 2 \bar 1 \bar 4 \bar 5} = \bar 3, 8, \bar 1, \bar 7, \bar 6, \bar 5, \bar 4, 2,10,9 \in \cK^\BC_5
\]
When $W \in \{S_n, \W_n\}$, the map $w \mapsto \underline w$ is a bijection $\{ w=w^{-1} \in W\} \to \cK.$
In type $\D$, 
if we set  $e(w) \coloneqq | \{ i \in [n]  : w(i) < -i\}|$ for $w \in \W_n$,
then the same map is a bijection
$%\be\label{dun-eq}
\{ w=w^{-1} \in \WD_n : e(w)\text{ is even}\} \sqcup
\{ w=w^{-1} \in \W_n \setminus \WD_n : e(w)\text{ is odd}\}
\to \cK_n^\D.
$%\ee


\end{remark}

Fix $z \in \cK$. Let 
$ \Des(z): =\{ s \in S : \ell(zs) < \ell(z)\}
$ and 
$
\Asc(z) \coloneqq\{ s \in S : \ell(zs) > \ell(z)\}
$
where $\ell$ is the length function of $S_{2n}$, $\W_{2n}$, or $\WD_{2n}$,
according to whether the Coxeter group $W$ is $S_n$, $\W_n$, or $\WD_n$,
respectively.
Next let
\[ \Des^=(z) \coloneqq \{ s \in S : sz=zs\}
\quand
\Asc^=(z) \coloneqq \{ s \in S : zsz \in \{ s_i : i > n\} \},\]
and define $\Des^< (z) $ and $ \Asc^< (z)$ to be the complements of 
$\Des^=(z)\sqcup \Asc^=(z)$
in $ \Des(z)$ and $\Asc(z)$, respectively.


Finally, define 
$\cM = \cM(W)$ and $\cN = \cN(W)$ to be the
 free $\LL$-modules with respective bases
  $\{M_z : z \in \cK \}$ and $ \{N_z : z \in \cK \}$.
  Our third main result is the following:


\begin{theorem}\label{main-thm2}
Assume $(W,S)$ is a classical finite Coxeter system of type $\A_{n-1}$, $\BC_n$, or $\D_n$ for some $n\geq 2$.
Then there is a unique $\H$-module structure on $\cM$ in which
\[
H_s  M_z = \begin{cases} 
M_{s  zs}  &\text{if }s \in \Asc^<(z) \\
M_{s  zs} + (x - x^{-1}) M_z &\text{if }s\in \Des^<(z) \\
xM_z & \text{if }s \in \Des^=(z) \\
-x^{-1} M_z & \text{if }s \in \Asc^=(z)
\end{cases}
\quad\text{for all $s \in S$ and $z \in \cK$},
\]
and there is a unique $\H$-module structure on $\cN$ in which
\[
H_s  N_z = \begin{cases} 
N_{s  zs}  &\text{if }s \in \Asc^<(z) \\
N_{s  zs} + (x - x^{-1}) N_z &\text{if }s\in \Des^<(z) \\
-x^{-1}N_z & \text{if }s \in \Des^=(z) \\
x N_z & \text{if }s \in \Asc^=(z)
\end{cases}
\quad\text{for all $s \in S$ and $z \in \cK$}.
\]
These $\H$-modules are multiplicity-free direct sums of irreducible submodules.
If
$(W,S)$ is not of type
 $ \D_{n}$
 with $n$ even,
then $\cM$ and $\cN$ are both Gelfand models
for $\H$.
\end{theorem}

The module $\cM(S_n)$ can be identified with the Gelfand model for $\H(S_n)$
that Adin, Postnikov, and Roichman study in \cite{APR2007}.
On the other hand, 
$\cN(W)$ for each group $W \in\{S_n,\W_n,\WD_n\}$ is a deformation
%\footnote{
%In the notation of \cite[\S1.2]{APR2007}, one has $q=x^2$ and $T_i = -x H_{s_i}$.
%On fixing $z \in \cG^\BC_{n} \supset \cG^\A_{n-1}$, define $w \in \W_n$ to have
% $w(i) = z(i)$ if $|z(i)| \leq n$,  $w(i) = i$ if $z(i) > n$, and $w(i) = -i$ if $z(i) < n$.
%Then an isomorphism from $\cM_\QQ(S_n)$ to the module $V_n$ in \cite[\S1.2]{APR2007}
%is  given by the linear map with $M_z \mapsto C_w$.
%Similarly, if one sets $x=1$, then
% the linear map sending $N_z \mapsto e_w$ 
%  is an isomorphism from $\cN(W)$ to the representation in \cite[\S2]{AraujoBratten}
%  for each $W \in\{S_n,\W_n,\WD_n\}$.
%} 
of the $W$-representation 
described by Araujo and Bratten in \cite{AraujoBratten}.
%
Theorem~\ref{main-thm2} implies that $\cM(S_n)$ and $\cN(S_n)$ are both abstractly isomorphic 
%after an appropriate extension of scalars)
to the type $\A_{n-1}$ version of the $\H$-module studied by Lusztig and Vogan in \cite{LV};
see \cite[\S5.3]{LV}. It is not clear how to write down these isomorphisms in a concrete way.
%
Our construction for types $\BC_n$ and $\D_{2n+1}$ 
partly resolves an open problem mentioned in \cite[\S7]{APR2010}.


\section{Gelfand $W$-graphs}


Let $p \mapsto \overline p$ denote the ring automorphism of $\LL$ sending $x \mapsto x^{-1}$.
A map $\phi : \cA \to \cB$ between $\LL$-modules is \emph{antilinear}
if $\phi(pa) = \overline p \cdot \phi(a)$ for all $a \in \cA$. 
%
%
For the Iwahori--Hecke algebra $\H$ of any Coxeter system $(W,S)$,
there is a unique antilinear map $\H \to \H$, written $H \mapsto \overline H$ and called the \emph{bar operator},
such that $\overline{H_w} = (H_{w^{-1}})^{-1}$ for all $w \in W$.
This map is a ring involution.

The famous \emph{Kazhdan--Lusztig basis} of $\H$ \cite{KL} consists of the unique elements $\underline H_w$ for $w \in W$ satisfying
$\overline{\underline H_w} = \underline H_w \in H_w + \sum_{ \ell(y) < \ell(w)} x^{-1} \ZZ[x^{-1}] H_y.$
Our third main result constructs analogous ``canonical bases'' for the modules $\cM$ and $\cN$
from Theorem~\ref{main-thm2}. 
%
An \emph{$\H$-compatible bar operator} on an $\H$-module $\cA$
is an antilinear map $A \mapsto\overline A$ with $\overline{HA} = \overline H \cdot \overline A$ 
for all $ H \in \H$ and $A \in \cA$.
For the rest of this section,
assume $(W,S)$ is a classical finite Coxeter system and define $\cK$, $\cM$, and $\cN$ as in Theorem~\ref{main-thm2}.

\begin{theorem}\label{main-thm3}
There are unique $\H$-compatible bar operators on $\cM$ and $\cN$
with
$\overline{M_z} = M_z$ and $\overline{N_z} = N_z$ for all $z \in \cK$ with 
$\Des^{<}(z) = \varnothing$.
Each of these bar operators is an involution.
%
Additionally, 
the module $\cM$ has a unique basis $\{ \underline M_z : z \in \cK\}$ satisfying
$\overline{\underline M_z} = \underline M_z \in M_z + \sum_{\substack{%y \in \cK \\ 
\ell(y) < \ell(z)}} x^{-1} \ZZ[x^{-1}] M_y.$
The module $\cN$ likewise has a unique basis 
$\{ \underline N_z : z \in \cK\}$ satisfying
$\overline{\underline N_z} = \underline N_z \in N_z + \sum_{\substack{%y \in \cK \\ 
\ell(y) < \ell(z)}} x^{-1} \ZZ[x^{-1}] N_y.$
\end{theorem}

The structure constants for multiplication $\H \times \cM \to \cM$ and $\H \times \cN \to \cN$
in the bases $\{ \underline M_z\}$ and $\{\underline N_z\}$
may be encoded using Kazhdan and Lusztig's notion of a \emph{$W$-graph} from \cite{KL}.
We review the definition below, following the conventions in \cite{Stembridge}.

\begin{definition}
An \emph{$S$-labeled graph}
  $\Gamma= (V,\omega,I)$ is a set $V$ with maps
 $\omega : V\times V \to \ZZ[x,x^{-1}]$
and $I : V \to \{\text{ subsets of $S$ }\}.$
%
We often think of this structure as a weighted directed graph on $V$
with edges $u\xrightarrow{\omega(u,v)} v$ for each $u,v \in V$ with $\omega(u,v) \neq 0$.
\end{definition}
\begin{definition}
  An $S$-labeled graph $\Gamma=(V,\omega,I)$ 
  is a \emph{$W$-graph}
if
   the free $\LL$-module $\cY(\Gamma)$ with basis $\{ Y_v : v \in V \}$ has an $\H$-module structure where
  \be\label{hy-eq}
    H_s Y_u = \begin{cases} x Y_u &\text{if }s \notin I(u) \\ 
 -x^{-1} Y_u +  \ds\sum_{\substack{v \in V\\ s\notin I(v)}} \omega(u,v) Y_v &\text{if }s \in I(u)
  \end{cases}
\quad\text{for all $s \in S$ and $ u \in V$.}\ee
\end{definition}
\begin{definition}
A $W$-graph  $\Gamma=(V,\omega,I)$ is \emph{quasi-admissible} 
  if (1) the corresponding directed graph is bipartite,
(2) we always have $\omega(u,v) \in \ZZ$,
(3)  we have
  $\omega(u,v) =0$ 
 whenever  $I(u)\subset I(v)$,
 and 
 (4) we have
$\omega(u,v) = \omega(v,u)$
  whenever
   $I(u)\not\subset I(v) \not\subset I(u)$.
A quasi-admissible $W$-graph is \emph{admissible}
if we always have $\omega(u,v) \in \NN$.
\end{definition}
%

%Assume $(W,S)$ is a classical finite Coxeter system and continue from Theorem~\ref{main-thm3}.
Define $\m_{yz}, \n_{yz} \in \ZZ[x^{-1}]$ for $y,z \in \cK$
to be the polynomials with
$\underline M_z = \sum_{y \in \cK} \m_{yz} M_y
$
and
$
\underline N_z = \sum_{y \in \cK} \n_{yz} N_y.$
Write
$ \mu^\m_{yz} \coloneqq [x^{-1}]  \m_{yz}$
and
$\mu^\n_{yz} \coloneqq [x^{-1}]  \n_{yz}$
for the coefficients of $x^{-1}$ in these polynomials. % $\m_{yz}$ and $\n_{yz}$.
For $z \in \cK$, define
\[\Asc^\m(z) \coloneqq \Asc^<(z) \sqcup \Asc^=(z)\quand \Asc^\n(z) \coloneqq \Asc^<(z) \sqcup \Des^=(z).\] 
Let
$\omega^\m(y,z) \coloneqq \mu^\m_{yz} + \mu^\m_{zy}$ if $\Asc^\m(y)\not\subset \Asc^\m(z)$
and $\omega^\m(y,z) \coloneqq 0$ otherwise for $y,z\in\cK$.
Likewise define
$\omega^\n(y,z) \coloneqq \mu^\n_{yz} + \mu^\n_{zy}$ if $\Asc^\n(y)\not\subset \Asc^\n(z)$
and $\omega^\n(y,z) \coloneqq 0$ otherwise.
%
Now let 
$\Gamma^\m=\Gamma^\m(W) \coloneqq (\cK,  \omega^\m, \Asc^\m)
$
and
$
\Gamma^\n=\Gamma^\n(W) \coloneqq (\cK,  \omega^\n, \Asc^\n).
$



\begin{theorem}\label{main-thm4}
%When $W$ is a finite classical Coxeter group,
The $S$-labeled graphs $\Gamma^\m$ and $\Gamma^\n$ are
quasi-admissible $W$-graphs. The linear maps $Y_z \mapsto \underline M_z$ and $Y_z \mapsto \underline N_z$ 
are isomorphisms $\cY(\Gamma^\m) \cong \cM$ and $\cY(\Gamma^\n)\cong \cN$.
\end{theorem}

These $W$-graphs are not usually admissible.
% the first negative edge weights occur when $W\in\{S_8,W^\BC_4,W^\D_7\}$.
%%and for all higher rank groups in each type.
%To actually compute 
%$\underline M_z$ and $\underline N_z$, one can use \eqref{hy-eq}, which becomes an inductive formula after substituting 
%$Y_z \mapsto \underline M_z$ and $Y_z \mapsto \underline N_z$.
%
Define a \emph{Gelfand $W$-graph} %for a finite Coxeter system $(W,S)$
to be a $W$-graph $\Gamma$ such that $\cY(\Gamma)$ is a Gelfand model for $\H(W)$.
Theorem~\ref{main-thm4} constructs a pair of Gelfand $W$-graphs $\Gamma^\m$ and $\Gamma^\n$
for the groups $W \in \{S_n ,\W_n, \WD_{2n+1}\}$. 
In the next section, we explain a precise sense in which these Gelfand $W$-graphs should be considered
as the canonical ones for classical types.


\section{Model equivalence}

Let $\TT = (J,z_{min},\sigma)$ be a model triple for a finite Coxeter system $(W,S)$.
Each map $\alpha \in \Aut(W,S)$ extends to an automorphism of $W^+$ that preserves $\sI(W,S)$  by the formula $\alpha : (w,\varphi) \mapsto (\alpha(w), \alpha \varphi\alpha^{-1})$. Using this convention, we define
\[ \TT^\alpha \coloneqq (\alpha(J), \alpha(z_{min}), \sigma \alpha^{-1})
\text{ for $\alpha \in \Aut(W,S)$}
\quand 
\overline{\TT} \coloneqq (J, z_{min}, \sigma \cdot \sgn).\]
Both
 $\TT^\alpha$ and  $\overline{\TT}$ are again model triples for $(W,S)$.
 Write $w_J$ for the longest element in $W_J$ and set $w_0 \coloneqq w_S$.
 If $z = (y,\theta) \in \sI_J$ for $y \in W_J$ and $\theta \in \Aut(W_J,J)$ then 
we set 
\be\label{wz-vee-eq}
z^\vee \coloneqq (y^\vee,\theta^\vee)\text{ where } \begin{cases} y^\vee \coloneqq w_0\cdot w_J \cdot  y\cdot  w_0,  \\ 
\theta^\vee \coloneqq \Ad(w_0)\circ \Ad(w_J)\circ  \theta \circ \Ad(w_0),
\end{cases}
\ee
writing $\Ad(g) : w \mapsto gwg^{-1}$ for the inner automorphism induced by $g \in W$.
One can check that $z^\vee \in \sI_{w_0 J w_0}$ % for $J^\vee \coloneqq w_0 J w_0$,
and that if $z$ is $W_J$-maximal then $z^\vee$ is $W_{w_0 J w_0}$-minimal.
Thus, if $z_{\max}$ is the unique $W_J$-maximal element  in the $W_J$-orbit of $z_{\min}$,
then 
\be \TT^\vee \coloneqq (w_0 J w_0, z_{\max}^\vee,\sigma \circ \Ad(w_0))\ee
is another model triple.
Finally, 
if $\TT' = (J',z_{\min}',\sigma')$
is a model triple such that 
$ J=J',$  $C_{W_J}(z_{\min}) = C_{W_J}(z'_{\min})$, and $\Res_{C_{W_J}(z_{\min})}^{W_J}(\sigma) = \Res_{C_{W_J}(z_{\min})}^{W_J}(\sigma')$,
then we write $\TT \equiv\TT'$.


\begin{definition}
Let $\sim$ be the transitive closure of the relation on model triples that has 
$\TT \sim \overline{\TT} \sim \TT^\vee$, $\TT \sim \TT^\alpha$ for all $\alpha \in \Aut(W,S)$,
and $\TT \sim \TT'$ whenever $\TT \equiv \TT'$.
Two perfect models $\sP$ and $\sP'$ for the same finite Coxeter group $W$ are \emph{equivalent}
if there is a bijection $ \sP \to \sP'$ such that if $ (J,z_{min},\sigma) \mapsto (J',z'_{min},\sigma')$ then $(J,z_{min},\sigma) \sim (J',z'_{min},\sigma')$.
\end{definition}

Let us explain what makes this is a sensible definition.
Results in \cite{Marberg,RV}
associate to any model triple $\TT = (J,z_{min},\sigma)$ a pair of 
 $W_J$-graphs with vertex set $W_J/C_{W_J}(z_{min})$.
Howlett and Yin's method of $W$-graph induction
\cite{HowlettYin,HowlettYin2}
transforms these structures into certain $W$-graphs, which we call $\Gamma^\m(\TT)$ and $\Gamma^\n(\TT)$,
with vertex set $W^J \times W_J/C_{W_J}(z_{min})$. Here $W^J$ is the set of minimal length left coset representatives of $W_J$ in $W$.


 



We do not have room here to fully explain these constructions (see \cite{MZ}).
 We comment, however, that the $W$-graphs in Theorem~\ref{main-thm4}
 can be identified with the disjoint unions of 
 $\Gamma^\m(\TT)$ and $\Gamma^\n(\TT)$ as $\TT$ ranges over  the  model triples 
 in Theorem~\ref{main-thm1b}, and our notion of equivalence for model triples 
 interacts nicely with  these smaller $W$-graphs: % $\Gamma^\m(\TT)$ and $\Gamma^\n(\TT)$:
\begin{itemize}
\item Replacing $\TT$ by $ \TT^\alpha$
corresponds to applying $\alpha$ to the vertices of  
$\Gamma^\m(\TT)$ and $\Gamma^\n(\TT)$.

\item It holds that $\Gamma^\m(\overline\TT) = \Gamma^\n(\TT)$ and $\Gamma^\n(\overline\TT) = \Gamma^\m(\TT)$.


\item If $\TT \equiv \TT'$  then $\Gamma^\m(\TT) \cong \Gamma^\m(\TT')$ and $\Gamma^\n(\TT) \cong \Gamma^\n(\TT')$.

\item The $W$-graph 
$\Gamma^\m(\TT^\vee)$ is
 dual to $\Gamma^\n(\TT)$, and $\Gamma^\n(\TT^\vee)$ is dual to $\Gamma^\m(\TT)$.

\end{itemize}
There is a precise notion of \emph{duality} for $W$-graphs \cite[Prop. 1.2]{Stembridge}, but
here it suffices to define this as meaning that the underlying directed graphs are anti-isomorphic,
that is, they are isomorphic after all of the directed edges in one graph are reversed. 

%Concerning the last item: if $\Gamma = (V,\omega,I)$ and
%$ \Gamma' = (V',  \omega',  I')$ are $W$-graphs with a 
% bijection $V \to V'$, written $v\mapsto v'$,
%then $\Gamma$ and $\Gamma'$ are \emph{dual} via $v\mapsto v'$ if 
% $\omega'(u',v') = \overline{\omega(v,u)}$ and $I'(v') = S \setminus I(v)$; see \cite[Prop. 1.2]{Stembridge}.


 The \emph{cells} of a $W$-graph $\Gamma = (V,\omega,I)$ are the subsets of $V$ that make up the
  strongly connected components in the associated directed graph. 
  If $C \subset V$ is a cell then 
  $\Gamma|_C \coloneqq (C, \omega|_{C\times C}, I|_C)$
   is itself a $W$-graph, and so defines a \emph{cell representation}
   $\cY(\Gamma|_C)$ of the Iwahori--Hecke algebra $\H$.
Identifying the cells in any particular $W$-graph
   is a natural problem of interest.
   %
The remarks above imply that if we can describe the cells in
$\Gamma^\m(\TT)$ and $\Gamma^\n(\TT)$
then we can also describe the cells in
$\Gamma^\m(\TT')$ and $\Gamma^\n(\TT')$
when $\TT\sim \TT'$.

 


Say that a model triple $(J,z_{min},\sigma)$ is \emph{strict} if 
$\sigma(s) = \sigma(t)$ whenever $s,t \in J$ have $st\neq ts$.
A perfect model is \emph{strict} if all of its model triples are strict.
This always holds if there are only odd edges in the Coxeter graph of $(W,S)$,
as happens in types $\A_n$ and $\D_n$.

  \begin{theorem}
 Suppose $(W,S)$ is an irreducible finite Coxeter system. 
 If $|S| \neq 3$
 then $(W,S)$ has at most one equivalence class of strict perfect models.
 Moreover, if $(W,S)$
  has a perfect model and 
  is not of type $\mathsf{I}_2(2n)$,
then $(W,S)$ also has a strict perfect model.
 \end{theorem}

\begin{remark*}
In type $\mathsf{I}_2(2n)$ there are no strict perfect models.
In type $\BC_n$ for $n\neq 3$,
there are only 4 perfect models. Two of these are strict and the others are  uninteresting ``refinements'' obtained by replacing a single strict model triple by a pair of non-strict ones.
\end{remark*}

To sum up, each perfect model gives rise to two Gelfand $W$-graphs,
and for equivalent models the cells in these $W$-graphs are essentially the same.
The Gelfand $W$-graphs for the classical groups in Theorem~\ref{main-thm4} are derived from representatives of a unique equivalence classes of (strict) perfect models.
In types $\BC_n$ and $\D_n$, there is one more property that underscores how the Gelfand $W$-graphs from Theorem~\ref{main-thm4}  are canonical:
%Recall the notation $\underline w $ for elements of $ \cK$ from Remark~\ref{notation-rmk}.
%For $w \in \W_{n}$, let $-w\in \W_{n}$ be the permutation $i \mapsto -w(i) = w(-i)$, and when $w=w^{-1}$ define $\iota_n^\BC(\underline w) = \underline{-w}$. This map is an involution of $\cK^\BC_n$.
%For $z \in \cK^\D_n$ define $\iota_n^\D(z) = \iota_n^\BC(z)$ 
%if $n + |\{ i \in [n] : |z(i)| > n\}|$ is divisible by $4$ and otherwise
%set $\iota_n^\D(z) = s_0\cdot \iota_n^\BC(z) \cdot s_0$ where $s_0=(-1,1)$.
%This is an involution of $\cK^\D_n$.

\begin{theorem}\label{bcd-dual-thm}
If $W \in \{W^\BC_n,W^\D_n\}$, then 
 $\Gamma^\m(W)$ and $\Gamma^\n(W)$
are dual in the sense of being anti-isomorphic as directed graphs (via an explicit bijection given in \cite{MZ}).
\end{theorem}

%\begin{theorem}\label{bc-dual-thm}
%The $W^\BC_n$-graphs $\Gamma^\m(W^\BC_n)$ and $\Gamma^\n(W^\BC_n)$
%are dual to each other via $\iota_n^\BC$.
%\end{theorem}
%
%
%
%\begin{theorem}\label{d-dual-thm}
%The $W^\D_n$-graphs $\Gamma^\m(W^\D_n)$ and $\Gamma^\n(W^\D_n)$
%are isomorphic after reversing the arrow of each edge.
%\end{theorem}

The analogous property for $\Gamma^\m(S_n)$ and $\Gamma^\n(S_n)$ seems to hold if and only if $n\leq 4$.

\section{Molecules in type A}

 It is an interesting open problem to identify the cells in the $W$-graphs 
 $\Gamma^\m(W)$ and $\Gamma^\n(W)$ for each classical Weyl group $W$. For $W \in \{\W_n, \WD_n\}$, it follows by Theorem~\ref{bcd-dual-thm} that the cells in $\Gamma^\m(W)$ determine the cells in $\Gamma^\n(W$ and vice-versa.
The cell classification problems for $\Gamma^\m(S_n)$ and $\Gamma^\n(S_n)$ are genuinely different.

  The \emph{molecules} in a $W$-graph $\Gamma = (V,\omega,I)$
  are the equivalence classes in $V$ under the
  transitive closure of the relation with $u \sim v$ if $\omega(u,v)\neq 0 \neq \omega(v,u)$.
%  If $C \subset V$ is a cell then 
%  $\Gamma(C) \coloneqq (C, \omega|_{C\times C}, I|_C)$
%   is itself a $W$-graph.
%   Each molecule is naturally viewed as an undirected graph and is a subset of exactly one cell.
   These subset  do not inherit a $W$-graph structure, but every cell is a disjoint union of molecules.   





 
 
 %\subsection{Cells in type A}

Computations suggest that there are fewer cells than molecules
 in $\Gamma^\m(\W_n)$ for all $n\geq 6$ and in $\Gamma^\m(\WD_n)$ for all $n\geq 4$.
 Moreover, the cell representations for both of these $W$-graphs are often reducible.
  In type $\A$, however, the following holds for at least $n \leq 10$:
  



\begin{conjecture}
Each molecule in $\Gamma^\m(S_n)$ and $\Gamma^\n(S_n)$
is a cell, and the cell representations associated to these $S_n$-graphs are all irreducible.
\end{conjecture}

As a first step toward proving this conjecture, 
 we describe the molecules in  $\Gamma^\m(S_n)$ and $\Gamma^\n(S_n)$.
 The \emph{(Young) diagram} of a partition $\lambda =(\lambda_1\geq \lambda_2 \geq \dots \geq \lambda_k>0)$
is the set $\D_\lambda = \{ (i,j) \in [k]\times \ZZ: 1 \leq j \leq \lambda_i\}$. For us, a \emph{tableau} of shape $\lambda$
is just a map $T : \D_\lambda \to \ZZ$, which we envision as an assignment of numbers
to some set of positions in a matrix. 
%A tableau is \emph{semistandard} if its rows are weakly increasing and its columns are strictly increasing.
A tableau is \emph{standard} if its rows and columns are strictly increasing and its entries
are the numbers $1,2,3,\dots,m$ for some $m\geq 0$ without any repetitions.


Suppose $T$ is a tableau and $a \in \ZZ$.
Define $T \fromRSK a$ to be the tableau formed by the familiar \emph{Schensted insertion} process.
 In detail, start by inserting $a$ into the first row of $T$. 
At each stage, a number $x$ is inserted into a row.
Let $y$ be 
the first entry in the row with $x< y$.
If no such $y$ exists then $x$ is added to the end of the row;
otherwise, $x$ replaces $y$ and  
$y$ is inserted into
the next row.
Once this process terminates, the result is $T\fromRSK a$.

 \begin{definition}
 Suppose $T$ is a tableau and  $a\leq b$ are integers. 
 Let  $T \fromMRSK (a,b)$ and $T \fromNRSK (a,b)$ be the tableaux given as follows.
 \ben


\item[(a)] %Next, let $T \fromMRSK (a,b)$ be the tableau given as follows.
If $a=b$ then  $T \fromMRSK (a,b) \coloneqq T\fromRSK a$.
If $a<b$ and $(i,j)$ is the box of $T \fromRSK a$   not in $T$,
then form $T \fromMRSK (a,b)$ from $T \fromRSK a$ by adding $b$ to the end of row $i+1$.

\item[(b)] 
If $a=b$ then  $T \fromNRSK (a,b)\coloneqq T\fromRSK a$.
If $a<b$ and $(i,j)$ is the box of $T \fromRSK a$ not in $T$,
then form $T \fromNRSK (a,b)$ from $T \fromRSK a$ by adding $b$ to the end of column $j+1$.

\een
 \end{definition}
 
 We refer to the algorithms constructing $T \fromMRSK a$  and $T \fromNRSK a$ as \emph{$\fkm$-insertion} and \emph{$\fkn$-insertion}.
 Our notion of $\fkm$-insertion is closely related to what Beissinger describes as \cite[Algorithm 3.1]{Beissinger}, while $\fkn$-insertion appears to be new.
  For example, we have
{\small \[ \ytab{
  2 & 3 
  \\4
} \fromRSK 2
   \  = \
    \ytab{
  2 & 3 
  \\4
} \fromMRSK (2,2)
   \  = \
    \ytab{
  2 & 3 
  \\4
} \fromNRSK (2,2)
   \  = \
\ytab{
   2 & 2 
   \\
   3\\
   4
} \]}
while
{\small \[
\ytab{
  2 & 3 
  \\4
} \fromMRSK (2,5)
   \  = \
\ytab{
   2 & 2  \\
   3 \\
   4 \\
   5
} 
\quand
\ytab{
  2 & 3 
  \\4
} \fromNRSK (2,5)
   \  = \
\ytab{
   2 & 2  \\
   3 & 5
    \\ 4
}. \]}

For  $w = w_1w_2\cdots w_n \in S_n$ let
$\PRSK(w) = \emptyset \fromRSK w_1 \fromRSK w_2 \fromRSK \cdots \fromRSK w_n.$
%This is a standard tableau whose shape is a partition of $n$.
The molecules in the $S_n$-graph associated to the type $\A_{n-1}$ Kazhdan--Lusztig basis (see \cite{KL}) are precisely the sets $\{ w \in S_n : \PRSK(w) = T \}$
as $T$ ranges over all standard tableaux with $n$ boxes. Each of these molecules turns out to be a cell 
with irreducible cell representation \cite{KL}.
 
We classify the molecules in   $\Gamma^\m(S_n)$ and $\Gamma^\n(S_n)$ in a related way.
 Fix  $z \in \cK^\A_{n-1}$. Let 
$
b_1 < b_2 <\dots <b_p
$
(respectively, $c_1 < c_2 < \dots <c_q$)
be the numbers $b \in [n]$ with $z(b) < b$ (respectively, the numbers $c \in [n]$ with $n < z(c)$)
and set $a_i = z(b_i)$. %Next let $c_1 < c_2 < \dots <c_q$ be the numbers $c \in [n]$ with $c \leq n < z(c)$.
Then define
\[
\ba \PMRSK(z) &=  \emptyset \fromMRSK (a_1,b_1) \fromMRSK\cdots \fromMRSK (a_p,b_p)
\fromMRSK (c_1,c_1) \fromMRSK \cdots \fromMRSK (c_q,c_q),
\\
\PNRSK(z) &=  \emptyset \fromNRSK (a_1,b_1) \fromNRSK\cdots \fromNRSK (a_p,b_p)
\fromNRSK (c_q,c_q) \fromNRSK \cdots \fromNRSK (c_1,c_1)
.
\ea\]
Let $\lambda_\fkm(z)$ and $\lambda_\fkn(z)$  be the   shapes of $\PMRSK(z)$ and $\PNRSK(z)$ for $z \in \cK^\A_{n-1}$.
Given a partition $\lambda$ of $n$, define $C^\m_\lambda = \{ z \in \cK^\A_{n-1} : \lambda_\fkm(z) = \lambda\}$ and
 $C^\n_\lambda = \{ z \in \cK^\A_{n-1} : \lambda_\fkn(z) = \lambda\}$.



\begin{example}
%Given $z =z^{-1} \in S_n$ with fixed points $c_1<c_2<\dots < c_q$ in $[n]$, let $\underline z \in \cK(S_n)$
%be the element that maps $c_i \mapsto n + i$ and $a \mapsto z(a)$ for $a \in [n] \setminus \{c_1,c_2,\dots,c_q\}$.
%For example,
%\[ \underline{2134} = (1,2)(3,5)(4,6)(7,8) \in S_8.\]
Suppose $\lambda=(3,1)$. Then, using the notation $\underline w $ from Remark~\ref{notation-rmk},
we have
$
C^\m_\lambda = \{ \underline{2134}, \underline{3214}, \underline{4231}\}
$
and 
$
C^\n_\lambda = \{ \underline{2134}, \underline{3214}, \underline{1324}\}
$
since 
{\footnotesize
\[ \PMRSK(\underline{2134})=\PNRSK(\underline{1324}) = \ytab{1 &  3 & 4 \\ 2},
\quad
\PMRSK(\underline{3214})=\PNRSK(\underline{3124}) = \ytab{1 &  2 & 4 \\ 3},
\quad
\PMRSK(\underline{4231})=\PNRSK(\underline{2134}) = \ytab{1 &  2 & 3 \\ 4}.
\]}
\end{example}

A theorem of Beissinger implies the following interesting property of $\PMRSK(z)$:

\begin{proposition}[{\cite[Thm. 3.1]{Beissinger}}]
If $z=z^{-1} \in S_n$ is fixed-point-free, then
$\PMRSK(\underline z) = \PRSK(z)$.
\end{proposition}

Here is our last main theorem, along with a more precise conjecture:

\begin{theorem}
As $\lambda$ ranges over all partitions of $n$, 
the sets $C^\m_\lambda$ and $C^\n_\lambda$ are all nonempty and give the 
distinct molecules in $\Gamma^\m(S_n)$ and $\Gamma^\n(S_n)$, respectively.
\end{theorem}

% We can also now give a more precise conjecture:

\begin{conjecture}
If $\lambda$ is any partition of $n$ then the molecules
$C^\m_\lambda$
and
$C^\n_\lambda$ in the respective $S_n$-graphs $  \Gamma^\m(S_n) $ and $ \Gamma^\n(S_n)$ are cells, and the representations associated to these cells
are both isomorphic to the usual (irreducible) Specht module $\mathbb{S}_\lambda$ of
$\H(S_n)$.
\end{conjecture}

  \printbibliography
% 
\end{document}
